

Missouri University of Science and Technology Scholars' Mine

Biological Sciences Faculty Research & Creative Works

**Biological Sciences** 

01 Jan 2024

# A Review Of Dielectric Barrier Discharge Cold Atmospheric Plasma For Surface Sterilization And Decontamination

Kolawole Adesina

Ta Chun Lin

Yue-Wern Huang Missouri University of Science and Technology, huangy@mst.edu

Marek Locmelis Missouri University of Science and Technology, locmelism@mst.edu

et. al. For a complete list of authors, see https://scholarsmine.mst.edu/biosci\_facwork/422

Follow this and additional works at: https://scholarsmine.mst.edu/biosci\_facwork

Part of the Aerospace Engineering Commons, Biology Commons, Geological Engineering Commons, Mechanical Engineering Commons, and the Petroleum Engineering Commons

# **Recommended Citation**

K. Adesina et al., "A Review Of Dielectric Barrier Discharge Cold Atmospheric Plasma For Surface Sterilization And Decontamination," *IEEE Transactions on Radiation and Plasma Medical Sciences*, Institute of Electrical and Electronics Engineers, Jan 2024. The definitive version is available at https://doi.org/10.1109/TRPMS.2024.3349571

This Article - Journal is brought to you for free and open access by Scholars' Mine. It has been accepted for inclusion in Biological Sciences Faculty Research & Creative Works by an authorized administrator of Scholars' Mine. This work is protected by U. S. Copyright Law. Unauthorized use including reproduction for redistribution requires the permission of the copyright holder. For more information, please contact scholarsmine@mst.edu.

# A Review of Dielectric Barrier Discharge Cold Atmospheric Plasma for Surface Sterilization and Decontamination

Kolawole Adesina, Ta-Chun Lin, Yue-Wern Huang, Marek Locmelis, and Daoru Han, Member, IEEE

Abstract-Numerous investigations have shown that non-equilibrium discharges at atmospheric pressure, also known as "cold atmospheric plasma" (CAP) are efficient to remove biological contaminants from surfaces of a variety of materials. Recently, CAP has quickly advanced as a technique for microbial cleaning, wound healing, and cancer therapy due to the chemical and biologically active radicals it produces, known collectively as reactive oxygen and nitrogen species (RONS). This article reviews studies pertaining to one of the atmospheric plasma sources known as Dielectric Barrier Discharge (DBD) which has been widely used to treat materials with microbes for sterilization, disinfection, and decontamination purposes. To advance research in cold atmospheric plasma applications, this review discusses various types and configurations of barrier discharge, the role played by reactive species and other DBD-CAP agents leading to its antimicrobial efficacy, a few collection of DBD-CAP past studies specifically on surface, and emerging applications of DBD-CAP technology. Our review showed that non-thermal/equilibrium plasma generated from DBD could sterilize or disinfect surface of materials without causing any thermal damage or environmental contamination.

*Index Terms*—Dielectric barrier discharge, cold atmospheric plasma, bacterial biofilm, decontamination, sterilization.

#### I. INTRODUCTION

**C**OLD Atmospheric Plasmas (CAP) are weakly ionized gases with charged species including electrons, positively/negatively charged ions, and neutral species (atomic and/or molecular radicals and

This work did not involve human subjects or animals in its research. Kolawole Adesina is a Graduate Research Assistant at the Department of Mechanical and Aerospace Engineering, Missouri University of Science and Technology, Rolla, MO 65409, USA.

Ta-Chun Lin is a Graduate Research Assistant at the Department of Biological Sciences, Missouri University of Science and Technology, Rolla, MO 65409, USA.

Yue-Wern Huang is with the Department of Biological Sciences, Missouri University of Science and Technology, Rolla, MO 65409, USA.

Marek Locmelis is with the Department of Geosciences and Geological and Petroleum Engineering, Missouri University of Science and Technology, Rolla, MO 65409, USA.

Daoru Han is with the Department of Mechanical and Aerospace Engineering, Missouri University of Science and Technology, Rolla, MO 65409, USA (e-mail: handao@mst.edu). non-radicals) under electric fields. [1]–[4].These plasmas, which are also known as low temperature nonequilibrium plasmas, can be produced at low pressures 0 between 100 Pa and 100 kPa, and even atmospheric pressure above 100 kPa.

1

Cold atmospheric plasmas generally have a gas temperature of less than 100 °C and an ionization level of less than 0.1%. Traditionally, the study of low temperature plasma discharges was primarily focused on two areas: (1) Low pressure discharges (< 1 Torr) mainly used in semiconductor industries for surface cleaning, etching, and deposition, and (2) High pressure discharges (up to 1 atm) with applications to ozonizers, electrostatic precipitators, and other devices [5].

Cleaning, sterilization, and decontamination using CAP have been investigated over the past 20 years [6]-[15]. For example, several CAP-based cleaning techniques were developed to treat heat-sensitive components of devices exposed to bio-hazards including food industry vessels [16], contaminated equipment in locations vulnerable to acts of warfare [17], and medical devices [18]. For these purposes, researchers have created various CAP devices including plasma jets, dielectric barrier discharges (DBD), glow discharges, corona discharges, using various sources including sinusoidal kHz AC [19], DC [20], pulsed DC [21], radio frequency (RF) [13], and microwave frequency (MW) [22] to better evaluate the inactivation efficacy and mechanism [23]. The process of inactivation heavily depends on a wide range of parameters including the types of plasma sources, properties of surfaces and environments, distances between the plasma and the treated surface, operational settings, gas composition and flow rate, treatment time, and the type of microorganism. As a result, it is challenging to comprehensively compare the efficiency of all CAP cleaning techniques.

Dielectric barrier discharge (DBD), also known as barrier discharge or silent discharge, constitutes distinct form of AC discharge. It generates a robust nonequilibrium plasma at atmospheric pressure and moderate gas temperatures. This type of discharge is usually achieved through an arrangement comprising two elec-

trodes, with atleast one of two of them being covered by a dielectric layer strategically placed along their current path between the metallic electrodes. The incorporation of insulating layers on or between the powered electrodes gives the unique method for establishing a non-equilibrium atmospheric pressure discharge. DBD relies on capacitive coupling, necessitating time-varying voltages to drive its operation [24].

Some of the major distinctions between DBD and other known classical discharges are; (i) separation of electrodes and discharge through a dielectric material which effectively prevents electrode corrosion and etching, (ii) DBD's inability to function with DC voltages due to capacitive coupling of the dielectric, thereby demanding alternating voltages to generate a displacement current, (iii) DBD maintenance without the need for gas feed and large surface area of exposure than plasma jets [25], (iv) manifestation of different discharge modes including homogeneous glow and filamentary glow discharges, and (v) distinctive combination of quasi-continuous and non-equilibrium unique behavior exhibited by DBD which has sparked broad interest in its applications and investigations [24]. Many DBD configurations are subjected to AC voltages or pulsed power ranging from 1 to 100 kV, with frequencies spanning through line frequency to several megahertz.

The active species from DBD-CAP that interact with either liquids or air can produce reactive species [26] include reactive oxygen (ROS) and nitrogen (RNS) species. Reactive oxygen and nitrogen species (RONS), such as 'OH, 'NO, 'NO<sub>2</sub>, 'O<sub>2</sub><sup>-</sup>, HNO<sub>3</sub>, H<sub>2</sub>O<sub>2</sub>, and N<sub>x</sub>O<sub>y</sub> has shown to play more important role in chemical reactions and physical modifications on solid surfaces, liquids, and soft matters (including cells and tissues) [27], [28]. For example, Yan et al. [29] investigated if atmospheric plasma can enhance microbial inactivation, wound healing, and cancer treatment of bone implant materials through the release of RONS. Further, to establish a baseline for clinical applications specifically in dental implants, these authors treated titanium with atmospheric low-temperature plasma. Their results show that treated titanium increased the capability of cells stickiness and growth on the surface of the metal.

Although, the precise mechanisms involved in actual inactivation pathways of cold plasma remain unclear [30]–[32], there is a growing consensus that the mechanism depends on the type of gases that are employed to create the plasma. It has also been shown that reactive chemical species, UV light, and electric fields all contribute to the antimicrobial effects of the low-temperature plasmas [1]–[3], [33], [34]. Depending on the techniques used in the plasma formation and whether the sample is subjected to direct or indirect plasma treatment, reactive chemical species may not always be the main inactiva-

tion factor to eliminate the target microorganism [35]. However, it is noted that the majority of investigations on the inactivation processes of plasma used bacterial spores. To fully understand the mechanisms driving the inactivation of different bacteria utilizing plasma, and the factors responsible for the decontamination, more research is needed [35].

The motivation for this review article is to offer comprehensive insight into past studies that utilized DBD for different applications related to the sterilization and decontamination of materials. The literature review focuses primarily on the use of dielectric barrier discharge due to its homogeneous, electrical insulation, and energy efficient nature used as a direct plasma source to sterilize material surfaces and perform microbial decontamination treatments. Atmospheric plasma from DBD has frequently been shown to possess high activity against microbes including multidrug resistant bacteria and fungi, on non-living surfaces, and in decontamination and sterilization of biofilms [36]-[41]. For instance, bacteria [42], [43], virus [44], [45], fungi [46], [47], and other pathogens [15] have been treated using atmospheric pressure plasma sources and a combination of inactivation agents i.e., reactive oxygen and nitrogen species, electric field, ultraviolet radiations e.t.c generated when in operation from which the results showed possible complete inactivation.

This review paper is organized as follows. Following the introduction, Section II describes the existing types of barrier discharges in a combination of quasicontinuous and non-equilibrium characteristics to generate atmospheric pressure discharge. Section III discusses various designs, configurations, and types of the DBD system and their working principles. Section III also describes previous studies that demonstrated successful sterilization of various materials and the decontamination of surface microbes using DBD technology. Section III-B reviews treated materials and biofilms, and discusses possibilities of disinfecting biotic as well as abiotic surfaces. Section V presents the technologies that are currently employed to examine surface morphology and to confirm the viability of microorganisms after treatment. Section VI describes the potential of cold plasma technologies in a range of emerging fields and/or applications. Finally, a summary and concluding remarks are given in Section VI.

# II. TYPES OF BARRIER DISCHARGES AND AFFILIATED DBD-CAP MICROBIAL INACTIVATION AGENTS

Typically, the DBD can manifest in two primary discharge modes: the filamentary mode, which represents the prevalent form characterized by numerous micro discharges distributed randomly across the surface of elec-

trode; or the homogeneous glow discharge, also referred to as the atmospheric pressure glow discharge mode due to its resemblance to DC glow discharge [48], [49]. A few types of DBD configuration currently used and the major DBD-CAP agents which may be responsible for the microbial sterilization and decontamination processes on surfaces are briefly discussed in this section.

### A. Types of Dielectric Barrier Discharge





Some major types of barrier discharges. (a) Cross-sectional view of a linear SDBD arrangement. (b) Linear SDBD arrangement having CAP along the border of the exposed electrode.
(c) Dielectric layers covering both electrodes in a VDBD setup.
(d) SDBD configuration with both electrodes embedded within the dielectric. (e) Typical plasma jet configuration. (f) Floating-electrode DBD (FE-DBD) with skin tissue as dielectric layer.

1) Surface DBD (SDBD): Surface discharge devices consist of a slender and long electrode positioned on a dielectric surface, accompanied by an elongated counter-electrode on the opposite side of the dielectric. Usually with this arrangement, a distinct discharge is not well-defined, allowing for propagation of discharge along the surface of the dielectric [50]. There is usually no gap between the electrodes and the dielectric barrier. Consequently, the discharges assume an almost two-dimensional distribution across the surface of the barrier and along the exposed edge of the electrode as illustrated in Figure 1(b). Also, the lower electrode is grounded and concealed beneath an additional layer of dielectric material to prevent any potential plasma formation on the side of the SDBD generator. This is clearly depicted in a cross-sectional

view in Figure 1(a). Alternatively, another SDBD reactor, shown in Figure 1(d), involves placing both electrodes side by side and covering them with a layer of dielectric material [51]. All of these setup results in the generation of cold plasma over the barrier without direct contact between the electrodes and the surrounding gas.

- 2) Volume DBD (VDBD): In VDBD, microdischarges are initiated within a three-dimensional region defined by the gap between the electrode, as depicted in Figure 1(c). For this case, either one or both electrodes can be coated with a dielectric material. Volume discharge can basically exhibit the planar or coaxial configurations. For the planar electrode setups, both of the electrodes run parallel, and either one or two dielectric barriers are invariably positioned either on the ground or ground electrode, on both electrodes, or between the two metal electrodes. On the other hand, the coaxial arrangements are also possible for the DBD, it involves one electrode encased within another, followed by atleast one or two dielectric barriers that are positioned either: (i) on the outer surface of the inner electrode or the inner surface of the outer electrode, (ii) on both electrodes facing each other, or (iii) in the gap between the two cylindrical electrodes [24].
- 3) Co-planar DBD: The co-planar discharge is simply the combination of both volume and surface discharge configurations as seen in the Figure 2 employed in plasma display panels. Co-planar discharge device features sets of long electrodes with opposite polarities embedded in the surface proximity around dielectric bulk [52], [53].
- 4) Floating Electrode DBD (FE-DBD): In the dielectric isolator FE-DBD system, the process involves bringing the electrode energy close to the surface to be treated at a distance known as the discharge gap, which is less than 3 mm [54]. FE-DBD system was also employed with flexible electrodes to establish the plasma's characteristics managed by adjusting both the dielectric properties and the voltage applied to the flexible powered electrode [25]. As the name suggests, the ground or buried electrode is removed as seen in Figure 1(f), and the high-voltage electrode is left unconnected, or "floating". Under the circumstances, the applied voltage alone isn't potent enough to induce breakdown. So, when an high dielectric constant object, such as skin tissue goes closely to the electrode, it gives rise to a substantial capacitance where

majority of the voltage are concentrated. This, in turn, generates a robust electric field within the gap that may result in breakdown and formation of atmospheric plasma discharge [55].

- 5) Micro-hollow DBD: Rather than employing the conventional parallel plate configuration of DBD, an integrated have been made featuring coaxial micro-hollow discharges. This approach involves stacking two metals meshes coated with ceramic material, allowing for scalability while maintaining uniform discharge across the electrode area [56]. An additional advantage of this arrangement is its capacity to facilitate gas flow perpendicular to the electrode plane. This configuration enables the effective flow of active product species from the discharge plasma to the surfaces being processed in close proximity to to the electrode surface.
- 6) DBD plasma jets (DBD jet): The DBD jet amalgamates dielectric barrier discharge (DBD) and atmospheric pressure plasma jet (APPJ) techniques, ensuring a consistent and homogeneous plasma treatment process [57]. It has a specific type of electrode that requires the introduction of an extra gas as shown in Figure 1(e). This gas is subsequently ejected as a low-temperature plasma jet into ambient air. Since the temperature involved remain below 40 °C, making it easier to come into contact with delicate substances, including tissues, without inducing thermal harm [58]. The discharge from DBD jet was initiated by a sinusoidal high voltage in the kHz range and sustained using helium gas directed into a dielectric capillary tube, with one end exposed to the surrounding atmospheric air. In essence, this phenomenon involves the advancement of streamers within a channel created by both a dielectric medium and a noble gas permeating the ambient air. The presence of nobles gas aids ionization due to higher ionization coefficient in comparison to air at the same reduced electric field conditions [59].

### B. DBD-CAP Agents responsible for the Mechanism of Microbial Inactivation

The role of CAP agents for the microorganism inactivation process has been the subject of several studies in recent years [3], [6], [7], [60]. An essential question is how DBD-CAP eliminates microorganisms and the role of CAP species in the process. Further, Dobrynin et al. [61] concluded that the interaction of living entities with plasma involves both positive and negative plasma ions. Consequently, the antibacterial effectiveness is believed to be caused collectively, by these active species [62].

Here, we discuss the contributions of five primary inactivation components from dielectric barrier cold atmospheric plasma below:

- Reactive oxygen and nitrogen species (RONS): RONS are a class of highly reactive molecules that consist of both radical species and non-radical oxidants containing oxygen and nitrogen. These species can arise as either primary products or metabolic byproducts [4]. It is also suggested that the hydroxyl-radical, 'OH, damages the outer structure of bacterial cells via oxidation process [63]–[66]. The lipid oxidation process involves lipid peroxidation, DNA damage, protein modification, and eventually may lead to cell death [32], [67]. Ozone (O<sub>3</sub>) also has same potent bactericidal effect as it interferes with cellular respiration [68].
- 2) Charged particles (electrons and ions): The charged particle may also be capable of contributing to bacteria inactivation and  $O_2^-$  may be able to play an important role in the inactivation mechanism as produced by a direct CAP [69]. Therefore, the electrostatic stress in charged particles of cold plasma which refers to the force or pressure experienced by the particles due to the electric fields present in the in CAP discharge is believed to aid microbial inactivation [23]. The bombardment of bacterial cells by charged particles for eradication of microorganisms is expected to be limited under specific conditions [23], [70]. However, the accumulation of these particles may damage the cell as a result of electrostatic stress [71], [72]. Due to their minor contribution in inactivation process, the vast majority of the studies conducted in this field have omitted to look into whether or not electrons and ions play any part in the process of bacterium inactivation [68].
- 3) Ultraviolet Radiation: Cold plasma produces UV radiation over a wide range of wavelengths, from 100 nm up to 280 nm (vacuum UV), depending on the working gas. Atmospheric pressures prevent the propagation of those wavelengths below 200 nm, even though they generate energetic photons [73], [74]. Many types of UV lights have been used to produce CAP species to disinfect the surfaces of certain equipment and devices. Such UV sources include low-pressure mercury lamps, medium-pressure mercury lamps, and excimer lamps. Several milliwatt-seconds of UV light per square centimeter at wavelengths between 220 and 280 nm (germicidal range) is needed to kill cells [3], [75]-[79]. UV photons also have a crucial role in the production of reactive oxygen species (ROS), which can lead to lipid peroxidation and harm the cell membrane [71]. In most

cases, target samples for sterilization are situated in a low-pressure vessel, utilizing argon or a N2/O2 gas combination, when a wavelength range of this kind is to be employed. Similarly, low-pressure plasmas also can generate UV dosages that are sufficient to render bacterial cells inactive [80]. The power density of UV radiation is less than 50  $\mu$ W/cm<sup>2</sup> in DBD-CAP, indicating a minor or no significant role in the inactivation mechanism [3], because most of the UV radiation generated gets reabsorbed in cold atmospheric plasma [81].

- 4) Electric field: A plasma discharge is generated between the electrodes when an electric field of adequate strength to maintain electron-ion pairs in the gas is establish. The ignition of this plasma discharge near the electric field prompt charge particles to accelerate towards their respective electrode. Both the intensity of the electric field and resultant forces produced, can be adjusted by modifying the spatial orientation of the electrodes [82]. A lot of studies have investigated the impact of agents such as reactive species, charged particles, and UV radiation on the antimicrobial effectiveness of DBD-CAP. However, the influence of electric field has relatively been less explored and studied. Furthermore, the charged particles and the electric field could have substantial impact on the DBD-CAP sterilization mechanism. The mechanism responsible for sterilizing and decontaminating surfaces through DBD-CAP is quite intricate, but often regarded as a synergistic outcome involving reactive species, charged particles, and UV radiation [23].
- 5) Heat: Heat does not play a major role in the inactivation of bacterial cells since non-equilibrium plasma do not generate high temperatures [3]. Temperature studies in various cold plasma sources revealed that the biological samples exposed to the plasma did not significantly raise their temperature [58]. Laroussi and Leipold [3] observed the rotational band structure of the 0-0 transition of the second positive system of nitrogen with stimulative spectra at various temperature. They calculated the gas temperature in their plasma discharge (DBD in air) and also measured the temperature in a sample that was placed 2 cm distant from the discharge using a thermocouple probe. They concluded that thermal impacts do not have a significant contribution because their observations revealed a negligible temperature increase.

### III. METHODS TO GENERATE ATMOSPHERIC PLASMA IN DBD

# A. Designs and Operation of DBD Cold Atmospheric Plasma Systems

The use of DBD treatment generates uniform surface modification [83]. Figure 2 shows different existing configurations of the dielectric barrier discharge which has been used by various researchers classified into planar, cylindrical, co-planar, and surface discharges. Importantly, the effectiveness of DBD plasmas is influenced by the design and operational parameters of the plasma sources [84].



Fig. 2 Basic electrode arrangements of DBD configurations.

In DBD, a secure configuration is made for producing CAP with at least one of two electrodes covered with a dielectric substance including glass, ceramics, silicon, quartz, and polymers because insulating materials can prevent plasma transfer into an arc discharge or spark [85], [86]. One of the electrodes is usually connected to the ground, while the other electrode is supplied with high voltage. The discharge current and arc generated are consequently restricted by the dielectric material around the high voltage electrode [87]. The gap distance between the two electrodes can range from millimeters to several centimeters apart [88], depending on the operating voltage, fed gas, and configuration adopted [7]. When the DBD is operated at atmospheric pressure and surrounded by air, it generates an electrical discharge between two electrodes. Figure 3 (a) demonstrates the schematic design of a DBD setup where Teflon and

quartz materials are used as the cover dielectric materials. And Figure 3 (b) shows an actual experiment replicating the setup.



Fig. 3

Typical DBD plasma setup is shown in a schematic design and picture. (a) demonstrate the schematic diagram of the normal DBD plasma and (b) Shows the actual experimental setup for DBD. Courtesy of Tripti Thapa Gupta. Used with permission [89]. Copyright 2018 IEEE.

# B. Utilization of DBD for Materials Sterilization and Microbial Decontamination

This review is centered on bacterial biofilms which have complex structures composed of bacterial cells closeup in a self-produced extra-cellular matrix. This matrix are known to contain proteins, polysaccharides, and DNA, giving protection and structural support. Table III-B below summarizes some CAP bacterial biofilm treatment outcomes specifically using various barrier discharge for surface sterilization and microbial decontamination since they are found on various surfaces, such as medical devices (implants, catheters), teeth, and natural environments. Numerous studies have been conducted using the DBD's direct application of plasma inactivation of a wide range of gram-negative and grampositive bacteria. Inactivation also occurs to bacterial biofilms, spores, and viruses [35], [89]–[95].

The inactivation effect is usually represented by decimal logarithm of the ratio between the number of viable cells observed in suspension before and after a treatment using DBD CAP [96]. For instance, say an initial bacteria of  $10^3$  were present before applying the DBD cold atmospheric plasma, upon a decrease of 2  $\log_{10}$  after treatment, the final bacteria count would be  $10^1$  representing a reduction up to 100 times.

Venezia et al. [104] used an afterglow PlasmaSol device (PlasmaSol Corporation) to sterilize bacteria. This device has a sterilizing container with an embedded plasma-generating electrode powered around  $30\pm1$  W. A combination of 1% ethylene, 50% oxygen, and 49% nitrogen at a flow rate of 1 L/min was utilized as the reagent gas. The study demonstrated that a 2 minutes exposure duration of *B. atrophaeus* to plasma achieved a decrease of 5 log<sub>10</sub>. In an either dry or moist environment, the number of bacteria decreased by roughly 5

 $log_{10}$  after just 10 minutes of treatment.

Eto et al. [105] used a setup to decontaminate samples kept in Tyvek wrapping (approximately  $40 \times 40$  mm). To accommodate any form of Tyvek packaging, they employed an electrode arrangement that adopted a flexible sheet-like structure and this device was used to sterilize the wrapped samples. The DBD was placed close to the Tyvek package by a stainless steel electrode grid at 2.5 kV and 5 kHz. Stainless steel bowl-shaped discs were contaminated and placed inside the Tyvek packing facing the discharge. An initial population of  $3 \times 10^6$ Geobacillus stearothermophilus spores were utilized as test organisms for various combinations of process gases (oxygen and nitrogen). The findings indicated that 15 minutes of exposure can result in complete inactivation of all bacteria, where 50% N<sub>2</sub> and 50% O<sub>2</sub> produced the most effective inactivation. The discharge was thought to be related to the production of ozone, which has concentration of 80 ppm and the reactive species that are produced were seen to be the most efficient for the inactivation.

Hahnel et al. [106] tested the distance effect on inactivation of microorganisms, using a surface dielectric barrier discharge SDBD in ambient air as shown in Figure 4(a). A unique electrode geometry was created while the mean gas temperature of the discharge was held under 300 K using 10 kV sinusoidal ignition voltage, 2 kHz frequency in pulsed mode with 1 Hz, and a maximum of 500 ms plasma on-time. *B. atrophaeus* spores was used at relative humidity to investigate the microbial killing. Their results suggested significant dependence on humidity, having an actual decrease factor of 5 log<sub>10</sub> after a 5 minute plasma treatment with a 300 ms on-time per second.



#### Fig. 4

Circuit board materials were used to make a variety of surface discharge electrodes as test strip. Courtesy of Marcel Hähnel and Katrin Oehmigen. Used with permissions (**a**) [106] (**b**) [107]. Copyright 2010 John Wiley and Sons.

Oehmigen et al. [107] also studied the effects of direct plasma treatment using the surface dielectric barrier discharge shown in Figure 4(b), examined in a solution where a surface DBD of a similar design (as in 4(a)) was made to fit into a petri dish having a 60 mm diame-

Table III-B: Relat	ed CAF	Plasma	Treatments	using	Dielectric	Barrier	Discharge	for	Materials	Sterilization	and
Microbial Decontamination											

Control Strain	Type of DBD-CAP and Parameter (Freq., Voltages, Operating Power, Applied gas, and Rate of Flow)	Inactivation rate	Surface for Biofilm Formation, and Time for Sterilization or Decontamination	Ref.
Staphylococcus aureus	FE-DBD: 0.13 W/cm <sup>2</sup> power and 120 V	All biofilms were steril- ized	Coverslip and 96 well plate for < 2min	[97]
Pseudomonas aeruginosa	DBD: 50 Hz and 120 kV	Biofilm reduced to unde- tectable level	96 well plate and Cover slips for 5 min	[98]
Streptococcus mutans and saliva multispecies	Volume DBD (VDBD): 40 kHz, 10 kV and Ar at 0.05 slm and Hollow DBD (HDBD):37.6 kHz, 8.4 kV and Ar & $Ar_{+O2}$ at 1 and 0.01 slm	5.38 log for S. mutans and 5.67 log for saliva surface biofilm	Titanium disc for 10 min	[99]
Pseudomonas aeruginosa and Staphylococcus epidermidis	Surface DBD (SBD): Structured electrode planar SDBD-A: (13 kV and 20 kHz) and a wire electrode SBD-B: (8 kV and 30 kHz with compressed air at 0.5 slm)	Reduction of 7.1 and 3.8 log using SBD-A and SBD-B in P. aeruginosa and reduction of 3.4 and 2.7 log using SBD-A and SBD-B in S. epidermidis	Polycarbonate disc plate for 10 min	[37]
Escherichia coli and Staphylococcus aureus	DBD: 31 kV peak-to-peak, 1.5 kHz, 10 $\mu$ s pulse, and power density of 0.29 W/cm <sup>2</sup>	Inactivation up to >95% of biofilms	Discs of implant ma- terials for 3 min	[95]
Candida albicans	VDBD: 40 kHz, 10 kV and 16 W power, and HDBD: 37.6 kHz, 9 kV and 9W power, Ar at 6 slm and Ar+(1%)O <sub>2</sub> at 0.06 slm	Reduction of 5 log	Titanium disc for 10 min	[66]
Stereptococcus mutans	DBD: 2 W/cm <sup>3</sup> power density, 580 kHz and He at 2L/min	98% eliminated	Tooth slices for 30 s	[100]
Escherichia coli and Grampositive Staphylococcus aureus	DBD: 50 Hz, 15 kV, 6 mm dis- charge gap, $O_2$ gas at 2 slpm	Relatively faster biphasic nanofiber treatment	Bombyx mori silk fibroin (BMSF) for 10–60 s periods	[101]
Escherichia coli, Cronobacter sakazakii, and Staphylococcus aureus	Underwater DBD 18 kV, 20 kHz, dry air at a rate of 5 L/min	$\sim$ 90% bacteria inactiva- tion at 5.50 log, 6.88 log, and 4.20 log CFU- coupon for E. coli, C. sakazakii, and S. aureus respectively	Stainless steel coupons for 90 min	[102]
Staphylococcus aureus and Escherichia coli	Air-based DBD: 7 kV, 22 kHz, and chamber volume of around 6200 $cm^3$ with diameter of 8 cm and thickness of 2 cm. Electrode dis- tance was set at 4 mm	Biofilms disruption up to 70% and 85% for S. au- reus and E. coli respec- tively	Biofilm in 6- well plates for 4 min	[90]
Bacillus subtilis	Open-air DBD: micro-discharge current pulses from $7 \times 10^5$ to 1.5 $\times 10^6$ for 9.6-10.8 kV & power density from 10-20 W/cm <sup>-3</sup>	Log reduction of 2.7 in (5.2 4.3) $\times$ 10 <sup>3</sup> CFU ml <sup>-1</sup> no of bacteria	Polymer foils treated for 180 s	[94]
Bacillus subtilis	Low-pressure (LPDBD): low AC HV below 10kV, 6 and 10mm gap distance, 28% and 25% at 50 and 30 kPa max. discharge rates	No surviving cells were observed at LPDBD 6mm gap	Glass beads surface for 4 min	[103]

ter. Escherichia coli and Staphylococcus aureus ranging from  $(10^6 - 10^8 \text{ cfu/mL})$  were thoroughly inactivated within 15 minutes, depending on the volume of the liquid sample. Their measurements of pH, nitrate, and nitrite concentrations after DBD plasma or NO gas treatment showed that the primary cause of liquid acidification was the formation of nitric acid due to reactive nitrogen species generated by the plasma.

Leipold et al. [9] attempted to disinfect some spinning cutting tools used for slicing purposes in meat industries. The configuration of their DBD is shown in Figure 5. The cutting tool is a 450 mm diameter disc with shallow cones that rotates at a period of 3.6 seconds per cycle. Two DBD were employed to the opposite sides of the cutter, which serves as the grounded electrode. One 100 × 100 mm electrode size of DBD is placed at a 2 mm to 4 mm distance to the surface of the circular knife. An initial concentration of  $5 \times 10^7$  CFU/ml of L. innocua (a biological indicator for L. monocytogenes) in 240  $\mu$ l was sprayed directly onto the knife. Inactivation increased by roughly 4-5  $\log_{10}$  between 68-170 s and 360 W; compared with the second DBD having an applied power of 1.8 W yielding a decrease of 3.5  $\log_{10}$  for a 300 s procedure but 37 s exposure time.



#### Fig. 5

Experimental setup sketch for plasma surface cleaning of cutting knives used for slicing in meat business. Courtesy of Frank Leipold. Used with permission [9]. Copyright 2010 Elsevier.

Schwabedissen et al. [108] used a different arrangement to disinfect items in a closed packaging as shown in Figure 6. They also employed a similar surface DBD where two powered electrodes operating in the kHz range are attached to the outside of the enclosed package, while a third electrode is connected to the outer electrodes located on the inside wall. A discharge between the inner electrode and the outer powered electrode produced approximately 2-4 W of discharge power at a peak-to-peak voltage of around 8 kV. Ozone, a disinfectant, was produced inside the enclosed packaging when used with ambient air. A paper strip containing *B. subtilis* was placed within a polystyrol petri dish to assess the microbicidal effectiveness of the PlasmaLabel discharge. A 4  $\log_{10}$  decrease after 10 minutes of ozone exposure was obtained.



#### Fig. 6

A schematic design of the PlasmaLabel idea showing the arrangement in a closed packaging. (1) an inner electrode, (4) two powered electrodes outer the package (2 and 3), and the sealed package (5). Courtesy of Axel Schwabedissen. Used with permission [108]. Copyright 2007 John Wiley and Sons.

Ercan et al. [109] also employed floating-electrode (FE-DBD) and reported that contrary to prior speculation, non-thermal plasma treated fluids retain their antimicrobial properties for a longer time (three months by delay time and two years by solution aging). A wide variety of multidrug-resistant bacteria and fungi pathogens, in both their planktonic and biofilm forms, were all inactivated by this fluid-mediated and plasmabased technology, which was quicker than the previously reported 15 min holding time (the time at which the plasma-treated liquid comes in contact with the suspended bacteria). The antimicrobial impact is attributed not only to change in pH, presence of  $H_2O_2$  or nitric acid, but also to the likely formation of supplementary species or products that are accountable for the potent biocidal outcome. Therefore, thorough investigations involving chemical analysis are necessary to validate the existence of these substances or their products.

In terms of power and frequency dependent effects for the antimicrobial inactivation, decrease of  $3-7 \log_{10}$  was observed with DBD plasma ranging 8-13 kV and 20-30 kHz [37], while a reduction of 5 log<sub>10</sub> was obtained

using a 8–10 kV and 37–40 kHz plasma source [66].

Collectively, CAP's effectiveness to eliminate bacteria is influenced by a number of factors including, 1) its method of production, exposure time, voltage, and frequency, 2) the properties of the target bacteria such as species type, growth stage, 3) the presence of a protective biofilm, and 4) some environmental conditions, such as, the composition of the surrounding material, humidity, and acidity. The amount of bacteria initially present also can be a factor that influences the efficacy of plasma treatment [32].

# IV. MATERIALS AND SURFACES TREATED BY DBD-CAP

One of the biological application of low-temperature atmospheric pressure plasmas was to eliminate of microorganisms on material surfaces. Due to it's ability to quickly and safely sterilize a variety of materials including glass, polymers, and human tissue, cold atmospheric plasma are the subject of ongoing studies for cleansing and sterilization. [7], [18], [110]. From the standpoint of contamination control, the in-package generation of plasma reactive species presents a distinct advantage by effectively reducing post-processing contamination and the occurrence of cross-contamination incidents [111].

There have been a need of novel technologies to properly sterilize heat-sensitive medical instruments [112]– [114], and to disinfect wounds. This led to enthusiasm in developing low-temperature plasma to sterilize, decontaminate, or disinfect both biotic and abiotic surfaces [10], [115]–[121]. CAP may be able to offer a safe and efficient approach to address the challenges associated with peri-implantitis treatment. CAP has proven to be efficient in the elimination of bacterial biofilms [14], [37], [39]–[41], [43], [66], [90], [95], [97]–[99], [122], [123].

Recent literature demonstrates that CAP can have positive effects on the skin. This implies that CAP technology has potential in the development of cosmetic treatments that can help prevent aging [124], and may also be effective therapy of various skin diseases [75], [125], [126]. Additionally, CAP has demonstrated antitumor effects, enhance stem cells, and also signals potential in enhancing immunotherapy for cancer patients [127]–[135].

### V. METHODS TO ASSESS EFFECTIVENESS OF DBD-CAP AFTER TREATMENT

Since the emergence of DBD cold atmospheric plasma, various investigations has been conducted to adequately analyze and interpret the surface modifications of treated materials [88], some of which have been briefly discussed here. A study showed no difference in the surface roughness as determined by a surface profilometer gauge after Ti discs has been treated with CAP and then subjected to different storage conditions. Contact angle and surface roughness measurements were taken at the end various of time intervals. Optical Emission Spectra (OES) analysis showed the presence of OH and NO as plasma-generated species. The contact angle measurements showed that the samples stored in saline retained their hydrophilicity for a longer duration compared to those stored in room conditions and an inert ambient, while, surface roughness measurements using a profilometer indicated no significant difference compared to the control group (untreated Ti) [136].

Proliferation tests are used to measure the plasmainduced microbicidal effectiveness. There exist many viability assays such as BacTiter-Glo and LIVE/DEADBacLight [7]. These tests entail labeling several bacterial compounds with fluorescent substances, which enables the fluorescence signals to distinguish between healthy and non-living bacteria [137].

Plasma diagnostics also play a crucial role in understanding CAP application on material surfaces. These diagnostic techniques entail gathering essential information about the plasma's properties and behavior, characterization, and optimization of plasma parameters. Typically, Oscilloscopes are employed to conduct the electrical measurements, while the UV-VIS-NIR spectroscopy ranging from 190–2200 nm may be used with a fiber optic spectrometer for their optical properties [129]. Additionally, electron microscopy can provide evidence of ruptured membranes or other surface morphological abnormalities [138] or using an atomic microscopy [106], [139], [140].

## VI. OTHER EMERGING APPLICATIONS OF DBD-CAP TECHNOLOGY

Floating electrode dielectric barrier discharge (FE-DBD), a DBD-based plasma generator have also been utilized in modern plasma medicine [141]. Viruses like SARS-CoV-2, which just produced the COVID-19 pandemic, has shown potential of being eliminated using DBD low-temperature plasma [142]–[144].

It is important to recognize that CAP, like that produced by DBDs, can also be a crucial tool to disinfect space probes looking for signs of life on distant planets and moons. Additionally, DBD-CAP may be employed in manned deep space, lengthy trips like in the expedition to Mars, to sanitize and disinfect astronauts' equipment like hand, head, and body wears that may come into contact with alien germs during such lengthy voyages [88].

Sterilization of food packaging is essential in keeping foods safe for consumption. Cold atmospheric plasma generated using DBD may be used as a final decontamination step to remove pesticides from fresh food products and their packaging [14], [16], [46], [143], [145]–[150].

#### VII. CONCLUSION AND FUTURE PERSPECTIVE

Cold atmospheric plasma is anticipated to be used as an antimicrobial device in the future. DBD-CAP is highly effective in inactivating multidrug-resistant pathogens and bacterium on biofilms. The inactivation outcomes vary among different types of CAP devices, working gases, treated materials etc. Thus, it is crucial to understand various parameters to effectively design and optimize a CAP device. The variety of operational gas and power input significantly affects effectiveness of CAP treatment, and the inactivation of free radicals and oxidants largely depend on high-voltage electric fields and frequency.

We have collated literature pertaining to the recent advancement in cold atmospheric plasma DBD for sterilization and decontamination of biological and nonbiological materials. Reactive nitrogen species, reactive oxygen species, electric field, and charged particles play a major role in the inactivation mechanism of DBD-CAPs, while UV radiation and heat plays a minor or no significant role, making the inactivation mechanisms quite complex. To generally validate DBD-CAP as antimicrobial tool, there is a growing need for uniformity in terms of characterization, optimization, and methodology which will be used as a baseline for future studies in this field.

The reviewed literature have amply demonstrated and proven that CAP, such as the one produced from a dielectric barrier discharge, is a promising technology in the treatment of material surfaces and in other emerging areas already discussed. Although, other CAP devices such as Radio Frequency [13], Corona Discharges [151], and CAP Jets [152] have also been tested to inactivate microbial growth over medical devices, hospital instrument surfaces, and other materials.

Collectively, DBD-CAP has the potential to become a highly effective antimicrobial technology for surface treatment, lessening the burden in the food, health and other sectors, and is expected to be used as antimicrobial therapy.

#### ACKNOWLEDGMENT

This Research was sponsored by the U.S. Army Combat Capabilities Development Command Analysis Center (DAC) and was accomplished under Cooperative Agreement Number W911NF-22-2-0183. The views and conclusions contained in this document are those of the authors and should not be interpreted as representing the official policies, either expressed or implied, of the U.S. Army Combat Capabilities Development Command Analysis Center or the U.S. Government. The U.S. Government is authorized to reproduce and distribute reprints for Government purposes notwithstanding any copyright notation herein. All authors declare that they have no known conflicts of interest in terms of competing financial interests or personal relationships that could have an influence or are relevant to the work reported in this paper.

10

#### REFERENCES

- X. Lu, G. Naidis, M. Laroussi, S. Reuter, D. Graves, and K. Ostrikov, "Reactive species in non-equilibrium atmosphericpressure plasmas: Generation, transport, and biological effects," *Physics Reports*, vol. 630, 04 2016.
- [2] M. Vandamme, E. Robert, S. Lerondel, V. Sarron, D. Ries, S. Dozias, J. Sobilo, D. Gosset, C. Kieda, B. Legrain, J. Pouvesle, and A. Pape, "Ros implication in a new antitumor strategy based on non-thermal plasma," *International Journal of Cancer*, vol. 130, p. null, 2012. [Online]. Available: https://www.semanticscholar.org/ paper/10ac24514788cec2986f9a763a604126424a4866
- [3] M. Laroussi and F. Leipold, "Evaluation of the roles of reactive species, heat, and uv radiation in the inactivation of bacterial cells by air plasmas at atmospheric pressure," *International Journal of Mass Spectrometry*, vol. 233, no. 1-3, pp. 81–86, apr 2004.
- [4] S. S. Al-Shehri, "Reactive oxygen and nitrogen species and innate immune response," *Biochimie*, vol. 181, pp. 52–64, feb 2021.
- [5] M. Laroussi, "1995–2005: A decade of innovation in low temperature plasma and its applications," *Plasma*, vol. 2, no. 3, pp. 360–368, aug 2019.
- [6] M. Moisan, J. Barbeau, S. Moreau, J. Pelletier, M. Tabrizian, and L. Yahia, "Low-temperature sterilization using gas plasmas: a review of the experiments and an analysis of the inactivation mechanisms," *International Journal of Pharmaceutics*, vol. 226, no. 1-2, pp. 1–21, sep 2001.
- [7] J. Ehlbeck, U. Schnabel, M. Polak, J. Winter, T. von Woedtke, R. Brandenburg, T. von dem Hagen, and K.-D. Weltmann, "Low temperature atmospheric pressure plasma sources for microbial decontamination," *Journal of Physics D: Applied Physics*, vol. 44, no. 1, p. 013002, dec 2010.
- [8] M. Moreau, N. Orange, and M. Feuilloley, "Non-thermal plasma technologies: New tools for bio-decontamination," *Biotechnol*ogy Advances, vol. 26, no. 6, pp. 610–617, nov 2008.
- [9] F. Leipold, Y. Kusano, F. Hansen, and T. Jacobsen, "Decontamination of a rotating cutting tool during operation by means of atmospheric pressure plasmas," *Food Control*, vol. 21, no. 8, pp. 1194–1198, aug 2010.
- [10] J. Birmingham and D. Hammerstrom, "Bacterial decontamination using ambient pressure nonthermal discharges," *IEEE Transactions on Plasma Science*, vol. 28, pp. 51– 55, 2000. [Online]. Available: https://www.semanticscholar.org/ paper/7c7ef386140882d223911e477135e5851bc2b322
- [11] D. Butscher, "Non-thermal plasma inactivation of microorganisms on granular food products," *Technology, medicine and applied sciences - Chemical engineering*, 2016. [Online]. Available: https://www.semanticscholar.org/ paper/aff8cace7b8351699812bd683079bf091aba1a37
- [12] Z. Machala, K Hensel, and Y. Akishev, "Plasma for bio-decontamination. medicine and food security," Conference proceedings, 2012. [On-Available: https://www.semanticscholar.org/paper/ linel 6bf93a4a8af88750ac92ecae7f3e9b7d25227919

- [13] H. C. Baxter, G. A. Campbell, A. G. Whittaker, A. C. Jones, A. Aitken, A. H. Simpson, M. Casey, L. Bountiff, L. Gibbard, and R. L. Baxter, "Elimination of transmissible spongiform encephalopathy infectivity and decontamination of surgical instruments by using radio-frequency gas-plasma treatment," *Journal* of General Virology, vol. 86, no. 8, pp. 2393–2399, aug 2005.
- [14] B. A. Niemira, G. Boyd, and J. Sites, "Cold plasma rapid decontamination of food contact surfaces contaminated with salmonella biofilms," *Journal of Food Science*, vol. 79, no. 5, pp. M917–M922, apr 2014.
- [15] B. Elmoualij, O. Thellin, S. Gofflot, E. Heinen, P. Levif, J. Seguin, M. Moisan, A. Leduc, J. Y. Barbeau, and W. Zorzi, "Decontamination of prions by the flowing afterglow of a reduced-pressure n2o2 cold-plasma," *Plasma Processes and Polymers*, vol. 9, pp. 612–618, 2012.
- [16] P. Kumar and J. Han, "Packaging materials for non-thermal processing of food and beverages," in *Emerging Food Packaging Technologies*. Elsevier, 2012, pp. 323–334.
- [17] K. Rastogi and L. Wallace, Handbook on Biological Warfare Preparedness. Environmental Sampling and BioDecontamination-Recent Progress, Challenges, and Future Direction. Elsevier, (2019), no. 195–208.
- [18] M. Mikhail and T. Young, "Sterilization of flexible endoscopes," in *Decontamination in Hospitals and Healthcare*. Elsevier, 2020, pp. 519–529.
- [19] H. B. Baniya, R. P. Guragain, B. Baniya, and D. P. Subedi, "Experimental study of cold atmospheric pressure plasma jet and its application in the surface modification of polypropylene," *Reviews of Adhesion and Adhesives*, vol. 8, no. 2, pp. S1–S14, jun 2020.
- [20] S. Wu, X. Lu, Z. Xiong, and Y. Pan, "A touchable pulsed air plasma plume driven by DC power supply," *IEEE Transactions* on *Plasma Science*, vol. 38, no. 12, pp. 3404–3408, dec 2010.
- [21] Y. L. Hu, Z. W. Du, and Z. Q. Chen, "Development of a pulsed DC power supply for generating cold plasma jet," Advanced Materials Research, vol. 791-793, pp. 1841–1844, sep 2013.
- [22] A. von Keudell, P. Awakowicz, J. Benedikt, V. Raballand, A. Yanguas-Gil, J. Opretzka, C. Flötgen, R. Reuter, L. Byelykh, H. Halfmann, K. Stapelmann, B. Denis, J. Wunderlich, P. Muranyi, F. Rossi, O. Kylián, N. Hasiwa, A. Ruiz, H. Rauscher, L. Sirghi, E. Comoy, C. Dehen, L. Challier, and J. P. Deslys, "Inactivation of bacteria and biomolecules by low-pressure plasma discharges," *Plasma Processes and Polymers*, vol. 7, no. 3-4, pp. 327–352, feb 2010.
- [23] S. Das, V. P. Gajula, S. Mohapatra, G. Singh, and S. Kar, "Role of cold atmospheric plasma in microbial inactivation and the factors affecting its efficacy," *Health Sciences Review*, vol. 4, p. 100037, sep 2022.
- [24] V. Nehra, A. Kumar, and H. Dwivedi, "Atmospheric non-thermal plasma sources," *International Journal of Engineering*, vol. 2, no. 1, pp. 53–68, 2008.
- [25] J.-H. Kim, J.-S. Park, Y.-S. Shin, and C.-K. Kim, "Atmosphericpressure floating electrode-dielectric barrier discharge with flexible electrodes: Effect of conductor shapes," *Korean Journal of Chemical Engineering*, vol. 36, no. 8, pp. 1371–1376, jul 2019.
- [26] D. Braný, D. Dvorská, J. Strnádel, T. Matáková, E. Halašová, and H. Škovierová, "Effect of cold atmospheric plasma on epigenetic changes, DNA damage, and possibilities for its use in synergistic cancer therapy," *International Journal of Molecular Sciences*, vol. 22, no. 22, p. 12252, nov 2021.
- [27] C.-Y. T. Tschang and M. Thoma, "In vitro comparison of direct plasma treatment and plasma activated water on escherichia coli using a surface micro-discharge," *Journal of Physics D: Applied Physics*, vol. 53, no. 5, p. 055201, nov 2019.
- [28] S. J. Kim and T. H. Chung, "Cold atmospheric plasma jetgenerated RONS and their selective effects on normal and carcinoma cells," *Scientific Reports*, vol. 6, no. 1, feb 2016.
- [29] M. Yan, P. Hartjen, M. Gosau, T. Vollkommer, A. L. C. Grust, S. Fuest, L. Kluwe, S. Burg, R. Smeets, and A. Henningsen, "Effects of a novel cold atmospheric plasma treatment of titanium on the proliferation and adhesion behavior of fibroblasts,"

International Journal of Molecular Sciences, vol. 23, no. 1, p. 420, dec 2021.

- [30] A. Shintani, H.; Sakudo, Gas Plasma Sterilization in Microbiology: Theory, Applications, Pitfalls and New Perspectives. Caister Academic Press, 2016.
- [31] P. B. Hideharu Shintani, Akikazu Sakudo and G. Mcdonnell, "Gas plasma sterilization of microorganisms and mechanisms of action," *Experimental and Therapeutic Medicine*, vol. 1, no. 5, pp. 731–738, 2010.
- [32] X. Liao, D. Liu, Q. Xiang, J. Ahn, S. Chen, X. Ye, and T. Ding, "Inactivation mechanisms of non-thermal plasma on microbes: A review," *Food Control*, vol. 75, pp. 83–91, may 2017.
- [33] D. B. Graves, "The emerging role of reactive oxygen and nitrogen species in redox biology and some implications for plasma applications to medicine and biology," *Journal of Physics D: Applied Physics*, vol. 45, no. 26, p. 263001, jun 2012.
- [34] E. Sardella, V. Veronico, R. Gristina, L. Grossi, S. Cosmai, M. Striccoli, M. Buttiglione, F. Fracassi, and P. Favia, "Plasma treated water solutions in cancer treatments: The contrasting role of rns," *Antioxidants*, vol. 10, p. null, 2021. [Online]. Available: https://www.semanticscholar.org/ paper/6c0c113e5629cdca89bac15ffc737742c7dbe42d
- [35] A. Sakudo, Y. Yagyu, and T. Onodera, "Disinfection and sterilization using plasma technology: Fundamentals and future perspectives for biological applications," *International Journal* of Molecular Sciences, vol. 20, no. 20, p. 5216, oct 2019.
- [36] M. Y. Alkawareek, Q. T. Algwari, G. Laverty, S. P. Gorman, W. G. Graham, D. O'Connell, and B. F. Gilmore, "Eradication of pseudomonas aeruginosa biofilms by atmospheric pressure non-thermal plasma," *PLoS ONE*, vol. 7, no. 8, p. e44289, aug 2012.
- [37] R. Matthes, C. Bender, R. Schlüter, I. Koban, R. Bussiahn, S. Reuter, J. Lademann, K.-D. Weltmann, and A. Kramer, "Antimicrobial efficacy of two surface barrier discharges with air plasma against in vitro biofilms," *PLoS ONE*, vol. 8, no. 7, p. e70462, jul 2013.
- [38] K. Fricke, I. Koban, H. Tresp, L. Jablonowski, K. Schröder, A. Kramer, K.-D. Weltmann, T. von Woedtke, and T. Kocher, "Atmospheric pressure plasma: A high-performance tool for the efficient removal of biofilms," *PLoS ONE*, vol. 7, no. 8, p. e42539, aug 2012.
- [39] A. J. Zelaya, G. Stough, N. Rad, K. Vandervoort, and G. Brelles-Marino, "Pseudomonas aeruginosa biofilm inactivation: Decreased cell culturability, adhesiveness to surfaces, and biofilm thickness upon high-pressure nonthermal plasma treatment," *IEEE Transactions on Plasma Science*, vol. 38, no. 12, pp. 3398–3403, dec 2010.
- [40] M. H. Lee, B. J. Park, S. C. Jin, D. Kim, I. Han, J. Kim, S. O. Hyun, K.-H. Chung, and J.-C. Park, "Removal and sterilization of biofilms and planktonic bacteria by microwaveinduced argon plasma at atmospheric pressure," *New Journal of Physics*, vol. 11, no. 11, p. 115022, nov 2009.
- [41] R. Sladek, S. Filoche, C. Sissons, and E. Stoffels, "Treatment of streptococcus mutans biofilms with a nonthermal atmospheric plasma," *Letters in Applied Microbiology*, vol. 45, no. 3, pp. 318–323, sep 2007.
- [42] J. W. Coburn and H. F. Winters, "Plasma etching—a discussion of mechanisms," *Journal of Vacuum Science and Technology*, vol. 16, no. 2, pp. 391–403, mar 1979.
- [43] K. G. Vandervoort and G. Brelles-Marino, "Plasma-mediated inactivation of pseudomonas aeruginosa biofilms grown on borosilicate surfaces under continuous culture system," *PLoS ONE*, vol. 9, no. 10, p. e108512, oct 2014.
- [44] A. Filipić, I. Gutierrez-Aguirre, G. Primc, M. Mozetič, and D. Dobnik, "Cold plasma, a new hope in the field of virus inactivation," *Trends in Biotechnology*, vol. 38, no. 11, pp. 1278–1291, nov 2020.
- [45] Z. Chen, G. Garcia, V. Arumugaswami, and R. E. Wirz, "Cold atmospheric plasma for SARS-CoV-2 inactivation," *Physics of Fluids*, vol. 32, no. 11, p. 111702, nov 2020.
- [46] N. Misra, B. Yadav, M. Roopesh, and C. Jo, "Cold plasma for

effective fungal and mycotoxin control in foods: Mechanisms, inactivation effects, and applications," *Comprehensive Reviews in Food Science and Food Safety*, vol. 18, no. 1, pp. 106–120, oct 2018.

- [47] N. Hojnik, M. Modic, Y. Ni, G. Filipivc, U. Cvelbar, and J. L. Walsh, "Effective fungal spore inactivation with an environmentally friendly approach based on atmospheric pressure air plasma," *Environmental Science & Technology*, vol. 53, pp. 1893 1904, 2019.
- [48] Y. M. T. Nozaki, Y. Unno and K.Okazaki., "A clear distinction of plasma structure between apg and dbd, paper," in 15th International Symposium on Plasma Chemistry, Orleans, France, 2001.
- [49] U. Kogelschatz, "Filamentary, patterned and diffuse barrier discharges," *IEEE Trans. Plasma Sci.*, vol. 30, pp. 1401–1407, 2002.
- [50] G. J. Pietsch and C. Humpert., "Discharge mechanism and ozone generation by surface discharges depending on polarity," in *In HAKONE 8th International Symposium on High Pressure Low Temperature Plasma Chemistry, Puhajarve Estonia*, 2002.
- [51] S. Portugal, B. Choudhury, and D. Cardenas, "Advances on aerodynamic actuation induced by surface dielectric barrier discharges," *Frontiers in Physics*, vol. 10, sep 2022.
- [52] L. Hulka and G. J. Pietsch, "On the ignition voltage and structure of coplanar barrier discharges," in *In HAKONE 8th International Symposium on High Pressure LowTemperature Plasma Chemistry, Puhajarve Estonia*, 2002.
- [53] T. M. V. I. Gibalov and G. J. Pietsch, "Parameters of barrier discharges in coplanararrangements," in *In HAKONE 8th International Symposium on High Pressure LowTemperature Plasma Chemistry, Puhajarve Estonia*, 2002.
- [54] G. Fridman, A. Shereshevsky, M. M. Jost, A. D. Brooks, A. Fridman, A. Gutsol, V. Vasilets, and G. Friedman, "Floating electrode dielectric barrier discharge plasma in air promoting apoptotic behavior in melanoma skin cancer cell lines," *Plasma Chemistry and Plasma Processing*, vol. 27, no. 2, pp. 163–176, mar 2007.
- [55] G. Fridman, M. Peddinghaus, M. Balasubramanian, H. Ayan, A. Fridman, A. Gutsol, and A. Brooks, "Blood coagulation and living tissue sterilization by floating-electrode dielectric barrier discharge in air," *Plasma Chemistry and Plasma Processing*, vol. 26, no. 4, pp. 425–442, jun 2006.
- [56] O. Sakai, Y. Kishimoto, and K. Tachibana, "Integrated coaxialhollow micro dielectric-barrier-discharges for a large-area plasma source operating at around atmospheric pressure," *Journal of Physics D: Applied Physics*, vol. 38, no. 3, pp. 431–441, jan 2005.
- [57] C.-Y. Shih, I.-C. Ni, C.-L. Chan, C.-C. Hsu, C.-I. Wu, I.-C. Cheng, and J.-Z. Chen, "Helium dielectric barrier discharge plasma jet (DBD jet)-processed graphite foil as current collector for paper-based fluidic aluminum-air batteries," *Energies*, vol. 15, no. 16, p. 5914, aug 2022.
- [58] M. Laroussi, "Cold plasma in medicine and healthcare: The new frontier in low temperature plasma applications," *Frontiers in Physics*, vol. 8, mar 2020.
- [59] P. Svarnas, P. K. Papadopoulos, D. Athanasopoulos, K. Sklias, K. Gazeli, and P. Vafeas, "Parametric study of thermal effects in a capillary dielectric-barrier discharge related to plasma jet production: Experiments and numerical modelling," *Journal of Applied Physics*, vol. 124, no. 6, aug 2018.
- [60] M. K. Boudam, M. Moisan, B. Saoudi, C. Popovici, N. Gherardi, and F. Massines, "Bacterial spore inactivation by atmospheric-pressure plasmas in the presence or absence of uv photons as obtained with the same gas mixture," *Journal of Physics D: Applied Physics*, vol. 39, no. 16, pp. 3494 – 3507, aug 2006.
- [61] D. Dobrynin, G. Fridman, G. Friedman, and A. Fridman, "Physical and biological mechanisms of direct plasma interaction with living tissue," *New Journal of Physics*, vol. 11, no. 11, p. 115020, nov 2009.
- [62] T. T. Gupta and H. Ayan, "Application of non-thermal plasma

on biofilm: A review," Applied Sciences, vol. 9, no. 17, p. 3548, aug 2019.

- [63] M. Laroussi and X. Lu, "Room-temperature atmospheric pressure plasma plume for biomedical applications," *Applied Physics Letters*, vol. 87, p. 113902, 2005. [Online]. Available: https://www.semanticscholar.org/paper/ 7dcb1f2ee859e1cc6b9c17cd3bcadbca7bbf66e8
- [64] G. Fridman, G. Friedman, A. Gutsol, A. B. Shekhter, V. N. Vasilets, and A. Fridman, "Applied plasma medicine," *Plasma Processes and Polymers*, vol. 5, no. 6, pp. 503–533, aug 2008.
- [65] M. Thiyagarajan, A. Sarani, and C. Nicula, "Optical emission spectroscopic diagnostics of a non-thermal atmospheric pressure helium-oxygen plasma jet for biomedical applications," *Journal* of Applied Physics, vol. 113, no. 23, p. 233302, jun 2013.
- [66] I. Koban, R. Matthes, N.-O. Hübner, A. Welk, P. Meisel, B. Holtfreter, R. Sietmann, E. Kindel, K.-D. Weltmann, A. Kramer, and T. Kocher, "Treatment of less candida albicans biofilms with low-temperature plasma induced by dielectric barrier discharge and atmospheric pressure plasma jet," *New Journal of Physics*, vol. 12, no. 7, p. 073039, jul 2010.
- [67] Kieft, IE (Ingrid), "Plasma needle:exploring biomedical applications of non-thermal plasmas," 2005.
- [68] M. Laroussi, "Nonthermal decontamination of biological media by atmospheric-pressure plasmas: review, analysis, and prospects," *IEEE Transactions on Plasma Science*, vol. 30, no. 4, pp. 1409–1415, aug 2002. [Online]. Available: https://www.semanticscholar.org/paper/ 9c535c45dd3f4ea32cd421cc876f0e27e9343bc3
- [69] X. Lu, T. Ye, Y. Cao, Z. Sun, Q. Xiong, Z. Tang, Z. Xiong, J. Hu, Z. Jiang, and Y. Pan, "The roles of the various plasma agents in the inactivation of bacteria," *Journal of Applied Physics*, vol. 104, no. 5, p. 053309, sep 2008.
- [70] K. D. Held, H. Kawamura, T. Kaminuma, A. E. S. Paz, Y. Yoshida, Q. Liu, H. Willers, and A. Takahashi, "Effects of charged particles on human tumor cells," *Frontiers in Oncology*, vol. 6, feb 2016.
- [71] N. O'Connor, O. Cahill, S. Daniels, S. Galvin, and H. Humphreys, "Cold atmospheric pressure plasma and decontamination. can it contribute to preventing hospital-acquired infections?" *Journal of Hospital Infection*, vol. 88, no. 2, pp. 59–65, oct 2014.
- [72] E. Stoffels, Y. Sakiyama, and D. B. Graves, "Cold atmospheric plasma: Charged species and their interactions with cells and tissues," *IEEE Transactions on Plasma Science*, vol. 36, no. 4, pp. 1441–1457, aug 2008.
- [73] J. Golda, B. Biskup, V. Layes, T. Winzer, and J. Benedikt, "Vacuum ultraviolet spectroscopy of cold atmospheric pressure plasma jets," *Plasma Processes and Polymers*, vol. 17, no. 6, p. 1900216, jan 2020.
- [74] G. Bruno, S. Wenske, H. Mahdikia, T. Gerling, T. von Woedtke, and K. Wende, "Radiation driven chemistry in biomolecules—is (v)UV involved in the bioactivity of argon jet plasmas?" *Frontiers in Physics*, vol. 9, dec 2021.
- [75] M. Hessling, R. Haag, N. Sieber, and P. Vatter, "The impact of far-uvc radiation (200–230 nm) on pathogens, cells, skin, and eyes – a collection and analysis of a hundred years of data," *GMS Hygiene and Infection Control; 16:Doc07*, 2021.
- [76] A. Sharma, A. Pruden, O. Stan, and G. Collins, "Bacterial inactivation using an RF-powered atmospheric pressure plasma," *IEEE Transactions on Plasma Science*, vol. 34, no. 4, pp. 1290– 1296, aug 2006.
- [77] H. Claus, "Ozone generation by ultraviolet lamps," *Photochemistry and Photobiology*, vol. 97, no. 3, pp. 471–476, feb 2021.
- [78] N. G. Reed, "The history of ultraviolet germicidal irradiation for air disinfection," *Public Health Reports*, vol. 125, no. 1, pp. 15–27, jan 2010.
- [79] M. Ploydaeng, N. Rajatanavin, and P. Rattanakaemakorn, "UVc light: A powerful technique for inactivating microorganisms and the related side effects to the skin," *Photodermatology*, *Photoimmunology & Photomedicine*, vol. 37, no. 1, pp. 12–19, oct 2020.

- [80] M. Laroussi, "Low-temperature plasmas for medicine?" *IEEE Transactions on Plasma Science*, vol. 37, no. 6, pp. 714–725, jun 2009.
- [81] X. Guimin, Z. Guanjun, S. Xingmin, M. Yue, W. Ning, and L. Yuan, "Bacteria inactivation using DBD plasma jet in atmospheric pressure argon," *Plasma Science and Technology*, vol. 11, no. 1, pp. 83–88, feb 2009.
- [82] M. Cheong, A. Greig, B. Gibson, and M. Arjomandi, "An investigation into the effect of electric field on the performance of dielectric barrier discharge plasma actuators," *Experimental Thermal and Fluid Science*, vol. 35, no. 8, pp. 1600–1607, nov 2011.
- [83] E. A. J. Bartis, A. J. Knoll, P. Luan, J. Seog, and G. S. Oehrlein, "On the interaction of cold atmospheric pressure plasma with surfaces of bio-molecules and model polymers," *Plasma Chemistry and Plasma Processing*, vol. 36, no. 1, pp. 121–149, oct 2015.
- [84] J. Laimer and H. Störi, "Recent advances in the research on non-equilibrium atmospheric pressure plasma jets," *Plasma Processes and Polymers*, vol. 4, no. 3, pp. 266–274, apr 2007.
- [85] P. J. Bruggeman, F. Iza, and R. Brandenburg, "Foundations of atmospheric pressure non-equilibrium plasmas," *Plasma Sources Science and Technology*, vol. 26, no. 12, p. 123002, nov 2017. [Online]. Available: https://dx.doi.org/10. 1088/1361-6595/aa97af
- [86] M. Laroussi and T. Akan, "Arc-free atmospheric pressure cold plasma jets: A review," *Plasma Processes and Polymers*, vol. 4, no. 9, pp. 777–788, nov 2007.
- [87] U. Kogelschatz, "Dielectric-barrier discharges: Their history, discharge physics, and industrial applications," *Plasma Chemistry and Plasma Processing*, vol. 23, no. 1, pp. 1–46, 2003.
- [88] M. Laroussi, "A brief note on the history of the dielectric barrier discharge and its application for biological decontamination," *IEEE Transactions on Radiation and Plasma Medical Sciences*, vol. 6, no. 1, pp. 121–125, jan 2022.
- [89] T. T. Gupta, J. S. Matson, and H. Ayan, "Antimicrobial effectiveness of regular dielectric- barrier discharge (dbd) and jet dbd on the viability of pseudomonas aeruginosa," *IEEE Transactions* on Radiation and Plasma Medical Sciences, vol. 2, no. 1, pp. 68–76, jan 2018.
- [90] S. Khosravi, S. Jafari, H. Zamani, and M. Nilkar, "Inactivation of staphylococcus aureus and escherichia coli biofilms by airbased atmospheric-pressure DBD plasma," *Applied Biochemistry and Biotechnology*, vol. 193, no. 11, pp. 3641–3650, aug 2021.
- [91] M. Laroussi, X. Lu, and M. Keidar, "Perspective: The physics, diagnostics, and applications of atmospheric pressure low temperature plasma sources used in plasma medicine," *Journal of Applied Physics*, vol. 122, no. 2, p. 020901, jul 2017.
- [92] P. Svarnas, A. Spiliopoulou, P. Koutsoukos, K. Gazeli, and E. Anastassiou, "Acinetobacter baumannii deactivation by means of DBD-based helium plasma jet," *Plasma*, vol. 2, no. 2, pp. 77–90, apr 2019.
- [93] M.Laroussi, "Low temperature plasma-based sterilization: Overview and state-of-the-art," *Plasma Processes and Polymers*, vol. 2, no. 5, pp. 391–400, jun 2005.
- [94] A. Kuzminova, T. Kretkova, O. Kylian, J. Hanus, I. Khalakhan, V. Prukner, E. Dolezalov, M. Simek, and H. Biederman, "Etching of polymers, proteins and bacterial spores by atmospheric pressure DBD plasma in air," *Journal of Physics D: Applied Physics*, vol. 50, no. 13, p. 135201, feb 2017.
- [95] F. Ibis, H. Oflaz, and U. K. Ercan, "Biofilm inactivation and prevention on common implant material surfaces by nonthermal DBD plasma treatment," *Plasma Medicine*, vol. 6, no. 1, pp. 33–45, 2016.
- [96] M. Beyrer, I. Smeu, D. Martinet, A. Howling, M. C. Pina-Pérez, and C. Ellert, "Cold atmospheric plasma inactivation of microbial spores compared on reference surfaces and powder particles," *Food and Bioprocess Technology*, vol. 13, no. 5, pp. 827–837, apr 2020.
- [97] S. G. Joshi, M. Paff, G. Friedman, G. Fridman, A. Fridman, and

A. D. Brooks, "Control of methicillin-resistant staphylococcus aureus in planktonic form and biofilms: A biocidal efficacy study of nonthermal dielectric-barrier discharge plasma," *American Journal of Infection Control*, vol. 38, no. 4, pp. 293–301, may 2010.

- [98] D. Ziuzina, S. Patil, P. Cullen, D. Boehm, and P. Bourke, "Dielectric barrier discharge atmospheric cold plasma for inactivation of pseudomonas aeruginosa biofilms," *Plasma Medicine*, vol. 4, no. 1-4, pp. 137–152, 2014.
- [99] I. Koban, B. Holtfreter, N.-O. Hübner, R. Matthes, R. Sietmann, E. Kindel, K.-D. Weltmann, A. Welk, A. Kramer, and T. Kocher, "Antimicrobial efficacy of non-thermal plasma in comparison to chlorhexidine against dental biofilms on titanium discs in vitro - proof of principle experiment," *Journal of Clinical Periodontology*, vol. 38, no. 10, pp. 956–965, jul 2011.
- [100] I. Molnar, J. Papp, A. Simon, and S. D. Anghel, "Deactivation of streptococcus mutans biofilms on a tooth surface using he dielectric barrier discharge at atmospheric pressure," *Plasma Science and Technology*, vol. 15, no. 6, pp. 535–541, jun 2013.
- [101] N. Ojah, D. Saikia, D. Gogoi, P. Baishya, G. A. Ahmed, A. Ramteke, and A. J. Choudhury, "Surface modification of core-shell silk/PVA nanofibers by oxygen dielectric barrier discharge plasma: Studies of physico-chemical properties and drug release behavior," *Applied Surface Science*, vol. 475, pp. 219–229, may 2019.
- [102] M. S. I. Khan, E.-J. Lee, and Y.-J. Kim, "A submerged dielectric barrier discharge plasma inactivation mechanism of biofilms produced by escherichia coli o157:h7, cronobacter sakazakii, and staphylococcus aureus," *Scientific Reports*, vol. 6, no. 1, nov 2016.
- [103] T. Tanino, M. Matsui, K. Uehara, and T. Ohshima, "Inactivation of bacillus subtilis spores on the surface of small spheres using low-pressure dielectric barrier discharge," *Food Control*, vol. 109, p. 106890, mar 2020.
- [104] R. A. Venezia, M. Orrico, E. Houston, S.-M. Yin, and Y. Y. Naumova, "Lethal activity of nonthermal plasma sterilization against microorganisms," *Infection Control and Hospital Epidemiology*, vol. 29, no. 5, pp. 430–436, may 2008.
- [105] H. Eto, Y. Ono, A. Ogino, and M. Nagatsu, "Low-temperature sterilization of wrapped materials using flexible sheet-type dielectric barrier discharge," *Applied Physics Letters*, vol. 93, no. 22, p. 221502, dec 2008.
- [106] M. Hahnel, T. von Woedtke, and K.-D. Weltmann, "Influence of the air humidity on the reduction of bacillus spores in a defined environment at atmospheric pressure using a dielectric barrier surface discharge," *Plasma Processes and Polymers*, vol. 7, no. 3-4, pp. 244–249, mar 2010.
- [107] K. Oehmigen, M. Hähnel, R. Brandenburg, C. Wilke, K.-D. Weltmann, and T. von Woedtke, "The role of acidification for antimicrobial activity of atmospheric pressure plasma in liquids," *Plasma Processes and Polymers*, vol. 7, no. 3-4, pp. 250–257, mar 2010.
- [108] A. Schwabedissen, P. Łaciński, X. Chen, and J. Engemann, "PlasmaLabel – a new method to disinfect goods inside a closed package using dielectric barrier discharges," *Contributions to Plasma Physics*, vol. 47, no. 7, pp. 551–558, nov 2007.
- [109] U. K. Ercan, H. Wang, H. Ji, G. Fridman, A. D. Brooks, and S. G. Joshi, "Nonequilibrium plasma-activated antimicrobial solutions are broad-spectrum and retain their efficacies for extended period of time," *Plasma Processes and Polymers*, vol. 10, no. 6, pp. 544–555, apr 2013.
- [110] E. Stoffels, A. A. Flikweert, W. Stoffels, and G. Kroesen, "Plasma needle: a non-destructive atmospheric plasma source for fine surface treatment of (bio)materials," *Plasma Sources Science and Technology*, vol. 11, pp. 383–388, 2002. [Online]. Available: https://www.semanticscholar.org/ paper/99d9dc57f16a2993332f44dbb27a385cddabacf3
- [111] P. Bourke, D. Ziuzina, L. Han, P. Cullen, and B. Gilmore, "Microbiological interactions with cold plasma," *Journal of Applied Microbiology*, vol. 123, no. 2, pp. 308–324, jun 2017.
- [112] A. L. Garner, A. M. Loveless, J. N. Dahal, and A. Venkattraman,

"A tutorial on theoretical and computational techniques for gas breakdown in microscale gaps," *IEEE Transactions on Plasma Science*, vol. 48, no. 4, pp. 808–824, apr 2020.

- [113] M. Moisan, K. Boudam, D. Carignan, D. Keroack, P. Levif, J. Barbeau, J. Seguin, K. Kutasi, B. Elmoualij, O. Thellin, and W. Zorzi, "Sterilization/disinfection of medical devices using plasma: the flowing afterglow of the reduced-pressure n 2 -o 2 discharge as the inactivating medium," *European Physical Journal-applied Physics*, vol. 63, p. 10001, 2013. [Online]. Available: https://www.semanticscholar.org/ paper/2f59aab53e23cdd3a874c9fb5c7097b890cffcba
- [114] M. Moisan, J. Barbeau, M.-C. Crevier, J. Pelletier, N. Philip, and B. Saoudi, "Plasma sterilization. methods and mechanisms," *Pure and Applied Chemistry*, vol. 74, pp. 349 – 358, 2002. [Online]. Available: https://www.semanticscholar.org/ paper/31e6316abb4ed26c1aff304ff19c44850f87edbf
- [115] M. Laroussi, "Low-temperature plasma jet for biomedical applications: A review," *IEEE Transactions on Plasma Science*, vol. 43, no. 3, pp. 703–712, mar 2015.
- [116] O. H. Chin, C. K. Lai, C. Y. Choo, C. S. Wong, R. M. Nor, and K. L. Thong, "Atmospheric pressure dielectric barrier discharges for sterilization and surface treatment," in *AIP Conference Proceedings*. AIP Publishing LLC, 2015.
- [117] N. D. Geyter and R. Morent, "Nonthermal plasma sterilization of living and nonliving surfaces." *Annual review of biomedical engineering*, vol. 14, pp. 255–74, 2012. [Online]. Available: https://www.semanticscholar.org/ paper/8e1106d08720d3fd6871128dbdcbf60bf4f85239
- [118] S. Lerouge, M. Wertheimer, and L. Yahia, "Plasma sterilization: A review of parameters, mechanisms, and limitations," *Plasmas and Polymers*, vol. 6, pp. 175–188, 2001. [Online]. Available: https://www.semanticscholar.org/ paper/cf5899c702c2c8ee12eb72080a24c0ccb3a7653c
- [119] K. Kelly-Wintenberg, T. Montie, C. Brickman, J. Roth, A. Carr, K. Sorge, L. Wadsworth, and P. Tsai, "Room temperature sterilization of surfaces and fabrics with a one atmosphere uniform glow discharge plasma," *Journal* of Industrial Microbiology and Biotechnology, vol. 20, pp. 69–74, 1998. [Online]. Available: https://www.semanticscholar. org/paper/3e4e520d337c866a2efcae21e4225b8c8c78797c
- [120] S. Moritz, A. Schmidt, J. Sann, and M. H. Thoma, "Surface modifications caused by cold atmospheric plasma sterilization treatment," *Journal of Physics D: Applied Physics*, vol. 53, no. 32, p. 325203, jun 2020.
- [121] F. Rossi and O. Kylián, "Sterilization and decontamination of surfaces by plasma discharges," 2012. [Online]. Available: https://www.semanticscholar.org/paper/ 3dd1a85f37a348683634c2bd279673124e6a5bdb
- [122] N. Hoiby, O. Ciofu, H. K. Johansen, Z. jun Song, C. Moser, P. O. Jensen, S. Molin, M. Givskov, T. Tolker-Nielsen, and T. Bjarnsholt, "The clinical impact of bacterial biofilms," *International Journal of Oral Science*, vol. 3, no. 2, pp. 55–65, apr 2011.
- [123] S. Pandit, M. VRSS, S. H. Helgadottir, F. Westerlund, and I. Mijakovic, "Combination of cold atmospheric plasma and vitamin c effectively disrupts bacterial biofilms," *Clinical Microbiology: Open Access*, vol. 06, no. 03, 2017.
- [124] G. Busco, E. Robert, N. Chettouh-Hammas, J.-M. Pouvesle, and C. Grillon, "The emerging potential of cold atmospheric plasma in skin biology," *Free Radical Biology and Medicine*, vol. 161, pp. 290–304, dec 2020.
- [125] L. Gan, J. Jiang, J. W. Duan, X. J. Z. Wu, S. Zhang, X. R. Duan, J. Q. Song, and H. X. Chen, "Cold atmospheric plasma applications in dermatology: A systematic review," *Journal of Biophotonics*, vol. 14, no. 3, dec 2020.
- [126] G. Fridman, "Direct plasma interaction with living tissue," 2008. [Online]. Available: https://www.semanticscholar.org/ paper/4d17a811da98ba2693058a9abf5f58b8268f9027
- [127] D. Brany, D. Dvorska, E. Halasova, and H. Skovierova, "Cold atmospheric plasma: A powerful tool for modern medicine," *International Journal of Molecular Sciences*, vol. 21, no. 8, p. 2932, apr 2020.

[128] D. Yan, J. H. Sherman, and M. Keidar, "The application of the cold atmospheric plasma-activated solutions in cancer treatment," *Anti-Cancer Agents in Medicinal Chemistry*, vol. 18, no. 6, pp. 769–775, nov 2018.

- [129] B. Trink, M. Keidar, J. Canady, Y. Shamai, and M. Tzukerman, "The effect of cold atmospheric plasma treatment on cancer stem cells," *Plasma Medicine*, 2015.
- [130] M. Keidar, A. Shashurin, O. Volotskova, M. Stepp, P. Srinivasan, A. Sandler, and B. Trink, "Cold atmospheric plasma in cancer therapy," *Physics of Plasmas*, vol. 20, p. 057101, 2012. [Online]. Available: https://www.semanticscholar.org/ paper/b3d6f5528530b3c643fcebb0d5de940f6b727743
- [131] D. Yan, J. Sherman, X. Cheng, E. Ratovitski, J. Canady, and M. Keidar, "Controlling plasma stimulated media in cancer treatment application," *Applied Physics Letters*, vol. 105, p. 224101, 2014. [Online]. Available: https://www.semanticscholar.org/paper/ 1f5688679ac2b61fba0a1d110b4f2c80189f69ac
- [132] F. Utsumi, H. Kajiyama, K. Nakamura, H. Tanaka, M. Mizuno, K. Ishikawa, H. Kondo, H. Kano, M. Hori, and F. Kikkawa, "Effect of indirect nonequilibrium atmospheric pressure plasma on anti-proliferative activity against chronic chemo-resistant ovarian cancer cells in vitro and in vivo," *PLoS ONE*, vol. 8, p. null, 2013. [Online]. Available: https://www.semanticscholar. org/paper/ddd47e648eade48df034e41898a5cd1b305cf239
- [133] H. Tanaka, S. Bekeschus, D. Yan, M. Hori, M. Keidar, and M. Laroussi, "Plasma-treated solutions (PTS) in cancer therapy," *Cancers*, vol. 13, no. 7, p. 1737, apr 2021. [Online]. Available: https://www.semanticscholar.org/ paper/79e527d3937ffbe9c584562d26ca2dd0f53c48e7
- [134] B. H. Adil, A. M. Al-Shammari, and H. H. Murbat, "Breast cancer treatment using cold atmospheric plasma generated by the FE-DBD scheme," *Clinical Plasma Medicine*, vol. 19-20, p. 100103, sep 2020.
- [135] R. Limanowski, D. Yan, L. Li, and M. Keidar, "Preclinical cold atmospheric plasma cancer treatment," *Cancers*, vol. 14, p. null, 2022. [Online]. Available: https://www.semanticscholar. org/paper/2829f176cc29d25ed677ed73efe70ddcef24f3a4
- [136] G. Yasav, G. D. Ekimci, M. Turkal, O. Karaman, and U. K. Ercan, "Evaluation of long term effect of cold atmospheric plasma on titanium implant materials," in 2019 Medical Technologies Congress (TIPTEKNO). IEEE, oct 2019.
- [137] C. Laflamme, S. Lavigne, J. Ho, and C. Duchaine, "Assessment of bacterial endospore viability with fluorescent dyes," *Journal* of Applied Microbiology, vol. 96, no. 4, pp. 684–692, apr 2004.
- [138] M. Laroussi, D. A. Mendis, and M. Rosenberg, "Plasma interaction with microbes," *New Journal of Physics*, vol. 5, pp. 41–41, apr 2003.
- [139] R. Pompl, F. Jamitzky, T. Shimizu, B. Steffes, W. Bunk, H.-U. Schmidt, M. Georgi, K. Ramrath, W. Stolz, R. W. Stark, T. Urayama, S. Fujii, and G. E. Morfill, "The effect of lowtemperature plasma on bacteria as observed by repeated AFM imaging," *New Journal of Physics*, vol. 11, no. 11, p. 115023, nov 2009.
- [140] S. P. Kuo, O. Tarasenko, S. Nourkbash, A. Bakhtina, and K. Levon, "Plasma effects on bacterial spores in a wet environment," *New Journal of Physics*, vol. 8, no. 3, pp. 41–41, mar 2006.
- [141] T. von Woedtke, S. Reuter, K. Masur, and K.-D. Weltmann, "Plasmas for medicine," *Physics Reports*, vol. 530, no. 4, pp. 291–320, sep 2013.
- [142] A. Bisag, P. Isabelli, G. Laghi, R. Laurita, G. Dirani, F. Taddei, C. Bucci, F. Capelli, M. Gherardi, A. Paglianti, V. Sambri, and V. Colombo, "Cold atmospheric plasma decontamination of SARS-CoV-2 bioaerosols," *Plasma Processes and Polymers*, vol. 19, no. 3, dec 2021.
- [143] M. Domonkos, P. Tichá, J. Trejbal, and P. Demo, "Applications of cold atmospheric pressure plasma technology in medicine, agriculture and food industry," *Applied Sciences*, vol. 11, no. 11, p. 4809, may 2021.
- [144] C. Labay, M. Roldan, F. Tampieri, A. Stancampiano, P. E. Bo-

canegra, M.-P. Ginebra, and C. Canal, "Enhanced generation of reactive species by cold plasma in gelatin solutions for selective cancer cell death," *ACS Applied Materials & Interfaces*, vol. 12, no. 42, pp. 47 256–47 269, oct 2020.

- [145] P. Peng, P. Chen, N. Zhou, C. Schiappacasse, Y. Cheng, D. Chen, M. Addy, Y. Zhang, E. Anderson, L. Fan, R. Hatzenbeller, Y. Liu, and R. Ruan, "Packed food and packaging materials disinfected by cold plasma," in *Advances in Cold Plasma Applications for Food Safety and Preservation*. Elsevier, 2020, pp. 269–286.
- [146] M. M. Nasiru, E. B. Frimpong, U. Muhammad, J. Qian, A. T. Mustapha, W. Yan, H. Zhuang, and J. Zhang, "Dielectric barrier discharge cold atmospheric plasma: Influence of processing parameters on microbial inactivation in meat and meat products," *Comprehensive Reviews in Food Science and Food Safety*, vol. 20, no. 3, pp. 2626–2659, apr 2021.
- [147] H. Tolouie, M. A. Mohammadifar, H. Ghomi, and M. Hashemi, "Cold atmospheric plasma manipulation of proteins in food systems," *Critical Reviews in Food Science and Nutrition*, vol. 58, no. 15, pp. 2583–2597, oct 2017.
- [148] O. Kylián, rei Choukourov, L. Hanyková, and H. Biederman, "Plasma technology for polymer food packaging materials," 2013. [Online]. Available: https://www.semanticscholar.org/ paper/adaf6987a7733ae7b9bb9df88d2849955dc8d08b
- [149] A. Frohling, "A flow cytometric approach to monitor the effects of gentle preservation techniques in the postharvest chain," 2011. [Online]. Available: https://www.semanticscholar. org/paper/c3b0673f41cd39c895691454835ae91914249d00
- [150] S. K. Pankaj and K. M. Keener, "Cold plasma: background, applications and current trends," *Current Opinion in Food Science*, vol. 16, pp. 49–52, aug 2017.
- [151] S. Kanazawa, T. Ikeda, F. Mitsugi, T. Ohkubo, and Y. Nomoto, "Control of DC corona discharge modes by a switching operation," *Plasma Processes and Polymers*, vol. 3, no. 9, pp. 692– 696, nov 2006.
- [152] M. Laroussi, "Low temperature plasma jets: Characterization and biomedical applications," *Plasma*, vol. 3, no. 2, pp. 54–58, apr 2020.



**Ta-Chun Lin** received a bachelor's degree in Life Science from National Taiwan Normal University, Taipei, Taiwan, in 2023, and is currently pursuing a master's degree in Biological Sciences at Missouri University of Science and Technology, Rolla, MO, USA. He served as an undergraduate research assistant at National Taiwan Normal University and has been working as a Graduate Research Assistant at Missouri University of Science and Technology since 2023. His research

15

interests include Alzheimer's disease, spinocerebellar ataxia type 17, and environmental toxicology.



Yue-Wern Huang received a bachelor's and master's degrees from National Taiwan Normal University in 1986 and 1988, respectively. He earned a Ph.D. from the University of Wisconsin-Madison in 1998. He joined the Missouri S&T faculty in 2000 as an assistant professor of biological sciences. He was named associate professor in 2006 and professor in 2012. Before joining Missouri University of Science and Technology, Rolla, MO, USA, he was a postdoctoral fellow at

Michigan State University from 1998-2000. He is the director of Missouri University of Science and Technology's Toxicology and Nanobiotechnology Laboratory and also chairs the Missouri S&T's Institutional Biosafety Committee (IBC). His current research interest include developing nanocarrier platforms to deliver biologically active molecules in vitro and in vivo for basic science research and biomedical applications/Investigate toxicity of nanomaterials to assess risk to human health; in particular, discover physical and chemical properties of nanomaterials that govern molecular mechanisms of nanotoxicity.



Kolawole Adesina received a bachelor's degree in Physics from Tai Solarin University of Education, Ogun, Nigeria, in 2019. He is currently pursuing a Ph.D. degree in Aerospace Engineering with the Missouri University of Science and Technology, Rolla, MO, USA. He was a Graduate Teaching Assistant at Tai Solarin University of Education from 2021 to 2022. He has been working as a Graduate Research Assistant since 2022 with the Missouri University of Science and Technology.

His research interests include low atmospheric plasma, plasma material interactions, environmental exposures, and plasma medicine.



Marek Locmelis received a bachelor's and master's degrees from Leibniz University in Hannover, Germany. He holds a Ph.D. in Earth and Planetary Sciences from Macquarie University in Sydney, Australia. He joined Missouri University of Science and Technology in 2016 from NASA Goddard Space Flight Center. Before NASA, he worked as a postdoctoral fellow at the University of Western Australia, Perth. He received a 2014 NASA postdoctoral fellowship award to in-

vestigate the evolution of redox conditions in Earths interior and a 2020 National Science Foundation (NSF) CAREER award for his research on translithospheric metal transport by magmatic-hydrothermal fluids. His current research interests include economic geology, planetary evolution, and analytical chemistry.



**Daoru Han** received a Bachelor of Engineering in Power Engineering of Aircraft (Aeronautical Propulsion) from Nanjing University of Aeronautics and Astronautics in 2005; a Master of Science in Aerospace Engineering degree with a thesis focused on aerodynamics from Missouri S&T; and a Ph.D. in Astronautical Engineering from the University of Southern California (USC), where he spent four years developing and applying firstprinciple-based plasma simulation models on

supercomputers to resolve fundamental plasma physics phenomena arising from space explorations. He is an Assistant Professor of Aerospace Engineering at Missouri University of Science and Technology. Previously, he was an Assistant Research Professor at Worcester Polytechnic Institute. His research interests include gas/plasma modeling, space propulsion, computational plasma physics, plasma dynamics, large-scale and high-performance computing, and plasma material interactions.